Author’s response to reviews

Title: The relationship between non-alcoholic fatty liver and skeletal muscle mass to visceral fat area ratio in women with type 2 diabetes

Authors:

Xiaoyou Su (sxy186028@163.com)
Jing Xu (xujing_6@126.com)
Chao Zheng (zhengchao_3@163.com)

Version: 1 Date: 25 Feb 2019

Author's response to reviews:

Dear Editor:

We wish to submit a manuscript for publication in BMC Endocrine Disorders, titled “The relationship between non-alcoholic fatty liver and skeletal muscle mass to visceral fat area ratio in women with type 2 diabetes”. The paper was coauthored by Xiaoyou Su, Jing Xu and Chao Zheng.

This manuscript has not been published or presented elsewhere in part or in entirety and is not under consideration by another journal. All authors have contributed significantly, and the study design was approved by the Institutional Review Board of the second affiliated hospital of Wenzhou Medical University and has been performed in accordance with ethical standards laid down in the Declaration of Helsinki. We have read and have abided by the statement of ethical standards for manuscripts submitted to the BMC Endocrine Disorders. If accepted, it will not be published elsewhere in the same form, in English or in any other language, including electronically without the written consent of the copyright-holder. There are no conflicts of interest to declare.
Thank you for your consideration. We look forward to hearing from you.

Sincerely,

Chao Zheng
Diabetes Center and Department of Endocrinology
2st Affiliated Hospital of Wenzhou Medical University
Lucheng District, Wenzhou
Zhejiang Province P. R. China
Tel& Fax: +86-15068256508
E-mail address: zhengchao_3@163.com